We think that the revisions enriched the manuscript

Here is point by point response:

Reviewer: Manuscript limitation should be indicated that the subject number is not enough to mentioned the conclusion.
Author response: Manuscript limitation has been added accordingly.

Reviewer: Authors are requested to send their revised manuscript to a professional English language editing company or a native English-speaking expert to polish the manuscript further.
Author response: The manuscript has been edited by a professional English language editing company.

Reviewer: In general, do not use non-standard abbreviations, unless they appear at least two times in the text preceding the first usage/definition
Author response: The abbreviations have been revised.